Author’s response to reviews

Title: Neutrophil CD64 index in cerebrospinal fluid as a marker of bacterial ventriculitis in children with external ventricular drainage

Authors:

Mojca Groselj-Grenc (mojcagg@gmail.com)
Andreja Natasa Kopitar (andreja-natasa.kopitar@mf.uni-lj.si)
Metka Derganc (metka.derganc@mf.uni-lj.si)
Maja Pavcnik (maja.pavcnik@kclj.si)

Version: 1 Date: 28 Nov 2017

Author’s response to reviews:

Technical Comments:

Please upload a clean revised manuscript that does not contain any tracked changes or highlighting as your main article file.

R: done

Editor's comments:

1) Please include a heading ‘Abstract’ in your manuscript.

R: done

2) Please amend your sub-heading ‘Introduction’ to ‘Background’ within your manuscript.

R: done

3) Please confirm in your ethics statement whether the participant consent was written or verbal. If it was verbal, please confirm that you received ethics committee approval for verbal consent and detail how you documented the consent.
Once amended, please also include a statement regarding ethics and consent in your methods section as well.

R: done

4) Please revise your current Availability of Data and Materials statement to reflect one of the formats indicated in our submission guidelines: https://www.biomedcentral.com/getpublished/editorial-policies#availability+of+data+and+materials

R: done

5) In your declarations statement 'Funding', please declare the role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

R: done

6) Please only provide figure titles/legends under a separate heading of 'Figure Legends' after the References and not within the text or figure files.

R: done

7) Please remove the duplicate copies of your tables that are uploaded as additional files. Tables should be included within your manuscript after your references, and your titles should be included above each table, not within the text.

R: done